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ABSTRACT 

 

Gasification of high-ash coal results in a syngas with large concentrations of various 

contaminants which necessitates a rigorous clean-up and conditioning process to produce 

syngas suitable for methanol production. As the design of the clean-up and conditioning 

process has a significant impact on the economy of the coal-to-methanol process, this work 

addresses intensification of the clean-up and conditioning process for syngas exiting a low-

grade coal gasifier. A model-based approach is followed, and numerical simulations of 

individual units are used to assess their performance.  

A one-dimensional model is developed to predict the performance of a pilot-scale 

bubbling fluidized bed gasifier for low-grade coal gasification. The model is found to predict 

the outlet composition of the pilot-scale gasifier with reasonable accuracy for different 

operating conditions. It is shown that the cold gas efficiency of the gasifier exhibits a maximum 

with the oxygen content of the fluidizing gas. It is also shown that the molar H2/CO ratio at the 

gasifier exit can be tailored by altering the steam feed rate. Further, an algorithm is proposed 

for screening and sequencing of syngas clean-up and conditioning processes. Using a sample 

syngas composition as the design basis, the feasibility of different sub-process schemes for 

clean-up and conditioning of syngas is examined by performing process simulations. 

Simulation of the entire clean-up and conditioning process scheme along with the closed-circuit 

methanol production unit is then performed and it is shown that the process scheme can yield 

methanol at 1000 kg/day. While the absorption-based clean-up units can be operated under 

varying load conditions using an over-design factor, the load-flexibility of the catalytic systems 

such as the partial water-gas shift unit needs to be ensured. Thus, a dynamic model is developed 

for a sour water-gas shift reactor and the optimum design and operating parameters are obtained 

by solving various optimization problems. The reactor designs where the molar H2/CO ratio at 
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the reactor exit is governed by equilibrium limitations are found to be dynamically more stable 

under fluctuating load. The model is extended to examine the possibility of carrying out two 

reactions: water-gas shift reaction and COS hydrolysis in a single fixed-bed reactor. Techno-

economic optimization studies are carried out to propose a heat-integrated multi-functional 

reactor design for water-gas shift and COS hydrolysis reactions. It is shown that low feed 

temperatures and high steam flow rates result in high COS conversion at a fixed CO 

conversion. However, reactor designs with higher feed temperatures eliminate the requirement 

of hot utility since the products leaving the reactor have sufficient enthalpy to pre-heat the feed 

syngas.  

Ab-initio calculations based on density functional theory are used to establish the 

dominant reaction pathway for COS hydrolysis on γ-Al2O3. Further, optimization of single-

pass and closed-circuit methanol production processes involving a multi-tubular reactor is 

carried out. Maximum COx conversion of 70% is obtained in the single-pass process while up 

to 90% COx conversion is obtained in the closed-circuit process. It is shown that the 

concentrations of inert species in the raw syngas significantly increases the reactor size and 

recycle processing costs in the closed-circuit process.  

The methodology and the results obtained in this work can be used for the design, 

analysis, and techno-economic optimization of a network of unit processes and unit operations 

with multiple heating and cooling operations. 
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सणरणंि  

उच्च-राख कोयले के गैसीकरण से उत्पन्ि ककए गए वायु, अर्ाात ‘ससिगैस’ मे अिेक दषूणकारी 

पदार्ा भारी मात्रा मे पाए जाते हैं। ऐसी ससिगैस से मेर्ेिॉल बिािे से पहले ससिगैस से इि दषूणकारी 

पदार्ों को निकाला जािा एवं ससिगैस के संघटि को मेर्ेिॉल संश्लेषण हेतु अिुकूल बिािा अनिवाया होता 

है। ससिगैस से दषूणकारी पदार्ों को निकालकर उसे मेर्िेॉल संश्लेषण हेतु अिुकूल बिािे वाली इस प्रकिया 

का कोयले से मेर्ेिॉल बिािे की सम्पूणा प्रकिया की लाभप्रदता पर काफी असर पड़ता है। इसीसलए प्रस्तुत 

प्रबंध मे कम दबाव वाले गैसीफायर से प्राप्त ककए गए ससिगैस से दषूणकारी पदार्ों को निकालकर उसे 

मेर्ेिॉल संश्लेषण हेतु अिुकूल बिािेवाली प्रकिया के गहिीकरण (इन्टेन्सफकैशि) को संबोधधत ककया गया 

है। प्रस्तुत संशोधि मे इस गहिीकरण हेतु गणणनतक प्रणाली का प्रयोग ककया गया है और प्रकिया के अलग-

अलग भागों के गणणनतक प्रनतरूपों की मदद से उि भागों की उपयोधगता जााँची गई है।   

इस संशोधि मे एक द्रववकृत गैसीफायर की उपयोधगता जााँचिे हेतु उस गैसीफायर का एक-आयामी 

गणणनतक प्रनतरूप निमााण ककया गया है। इस गणणनतक प्रनतरूप से अिुमानित ससिगसै का संघटि, प्रत्यक्ष 

मापे गए संघटि से उधचत मेल खाता है। प्रनतरूप की मदद से यह ददखाया गया है की गैसीफायर मे प्रवेसशत 

ककए गए वायु म ेऑक्सीजि के प्रमाण को मध्यम मूल्य पर रखिे से निसमात ससिगसै की ऊजाा-निसमाती 

क्षमता को अधधकतम ककया जा सकता है। तत्पश्चात यह भी ददखाया गया है की गैसीफायर मे प्रवेसशत ककए 

गए वायु मे भांप की मात्रा को बदलकर निसमात ससिगैस मे हाइड्रोजि और काबाि मोिोक्साइड की मात्राओं 

को संतुसलत ककया जा सकता है।  

ससिगैस से दषूणकारी पदार्ों को निकालकर उसे मेर्ेिॉल संश्लेषण हेतु अिुकूल बिािेवाली प्रकिया 

के घटको को छााँटिे हेतु तर्ा उन्हे िसमत करिे हेतु एक कलि ववधध (ऐल्गररदम) का निमााण इस संशोधि 
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मे ककया गया है। प्रकिया के घटकों के ववववध पयााय और उिकी उपयोधगता, उि पयाायों के गणणनतक 

प्रनतरूपों की मदद से जााँचे और चिुे गए है। ससिगसै स ेमेर्ेिॉल बिाि ेकी सम्पूणा प्रकिया का गणणनतक 

प्रनतरूप बिाकर यह ददखाया गया है की चिुे हुए पयाायों से बिी हुई प्रकिया से प्रनतददि १००० ककलोग्राम 

मेर्ेिॉल का उत्पादि हो सकता है। 

इस प्रकिया मे मौजूद संपे्रररत ररएक्टरों (catalytic reactors) का गैसीफायर से प्राप्त हुए ससिगैस 

के संघटि की अनियसमतता के प्रनत असंवेदिशील होिा जरूरी है। इसीसलए इस संशोधि मे प्रकिया के एक 

अहम घटक: ‘सार वाटर-गैस सशफ्ट’ ररएक्टर के सलए एक गनतशील गणणनतक प्रनतरूप (dynamic model) 

का निमााण ककया गया है और उस प्रनतरूप की मदद से ररएक्टर की रचिा और संचालि (डडजाइि) का 

अिुकूलि ककया गया है। यह ददखाया गया है की जजि संरचिाओ मे ररएक्टर से निकाससत वायु का संघटि 

रासायनिक संतुलि (chemical equilibrium) पर निभार करता है, ररएक्टर की वह संरचिाए गैसीफायर स े

प्राप्त हुए ससिगैस के संघटि की अनियसमतता के प्रनत असंवेदिशील होती है। इस गणणनतक प्रनतरूप को 

और ववकससत करके एक ऐसे ररएक्टर सरंचिा का संशोधि ककया गया है जजसमे दो अलग संपे्रररत प्रकियाए: 

‘सार वाटर-गैस सशफ्ट’ और काबोिील सल्फाइड (COS) के जलपे्रररत ववघटि (hydrolysis) की जा सके। 

इस संशोधि मे यह पाया गया है की ररएक्टर मे प्रवेसशत वायु के तापमाि को ववसशष्ट सीमा तक कम 

करके, और ररएक्टर मे भांप अधधक मात्रा मे प्रवेसशत करके काबोिील सल्फाइड का पररवताि स्तर बढ़ाया 

जा सकता है। परंतु, तकिीकी एवं आधर्ाक अिुकूलि को मध्यिजर रखके यह भी पाया गया है की ररएक्टर 

मे प्रवेसशत वायु के तापमाि को बढ़ा कर एक ऐसा ररएक्टर संरधचत ककया जा सकता है जहा ररएक्टर स े

निकाससत वायु, ररएक्टर मे प्रवेसशत वायु का तापमाि उधचत मात्रा तक बढ़ा सके।  
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‘डेजन्सटी फंक्शिल थ्योरी’ पर आधाररत गणिाओं के आधार पर, गामा अल्युसमिा पर काबोिील 

सल्फाइड (COS) के जलपे्रररत ववघटि (hydrolysis) मे अणु-स्तर पर होिे वाली प्रकियाओं का अभ्यास भी 

इस संशोधि मे ककया गया है। मरे्ेिॉल सशं्लेषण के सलए उपयोग ककए जािे वाले बहुिसलका ररएक्टर की 

दो अलग संरचिाओं का अिुकूलि भी इस संशोधि मे ककया गया है। पहली संरचिा मे ररएक्टर से निकाससत 

गैस पुिः चकित िहीं की जाती, और गैस मे मौजूद ७०% काबाि आक्साइडस ्(COx), मेर्िेॉल मे पररवनतात 

होती है। दसूरी संरचिा मे ररएक्टर से निकाससत गैस का आंसशक भाग ररएक्टर मे पुिः चकित ककया जाता 

है और गैस मे मौजूद ९०% काबाि आक्साइडस ्(COx), मेर्ेिॉल मे पररवनतात होती है।   

इस संशोधि मे प्रस्तुत गणणनतक प्रनतरूप एवं निरीक्षण ऐसी प्रकियाओं के महाजालों की संरचिा 

एवं अिुकूलि मे मददगार साबबत हो सकते है जहा एकाधधक ऊष्मा-ववनिमायक आवश्यक हो। 
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